

## **Author's response to reviews**

**Title:** Comparison of the adaptive implementation and evaluation of the Meeting Centers Support Program for people with dementia and their family carers in Europe; study protocol of the MEETINGDEM project

### **Authors:**

Rose-Marie Dröes (rm.droes@vumc.nl)

Franka Meiland (fj.meiland@vumc.nl)

Simon Evans (simon.evans@worc.ac.uk)

Dawn Brooker (d.brooker@worc.ac.uk)

Elizabetta Farina (efarina@dongnocchi.it)

Dorota Szcześniak (deren.dorota@gmail.com)

Lisa van Mierlo (lisa\_vanmierlo@hotmail.com;l.vanmierlo@vumc.nl)

Martin Orrell (m.orrell@nottingham.ac.uk)

Joanna Rymaszewska (Joanna.rymaszewska@umed.wroc.pl)

Rabih Chattat (rabih.chattat@unibo.it)

**Version: 1 Date:** 09 Mar 2017

### **Author's response to reviews:**

Dear Editor,

Please find attached the revised Manuscript including details about the type of consent obtained from participants.

Thank you and we hope this manuscript is now suitable for publication.

Best regards,

dr. Lisa van Mierlo